Author's response to reviews

Title: Rationale and design: telephone-delivered behavioral skills interventions for blacks with type 2 diabetes

Authors:

Leonard E Egede (egedel@musc.edu)
Joni L Strom (stromjl@musc.edu)
Valerie L Durkalski (durkalsv@musc.edu)
Patrick D Mauldin (mauldinp@musc.edu)
William P Moran (moranw@musc.edu)

Version: 4 Date: 24 March 2010

Author's response to reviews:

We have included reference to Figure 1 in the text of the manuscript as requested. See “study instruments and data collection schedule” on page 14 of the revised manuscript. Thanks.